\begin{table}%T6
\caption{\label{Tnlte3d}Estimated effects of NLTE~and 3D~on abundances from literature and adapted to the parameters of HE~0338$-$3945. The numbers are in many cases highly uncertain; see text for further discussion.} 
%\centerline
{
\begin{tabular}{llcrcccc}
\hline
\hline
\multicolumn{2}{l}{Element} & &
$\Delta_{{\rm NLTE}}$ & 
Ref.$_{{\rm NLTE}}$ & &
$\Delta_{{\rm 3D}}$ &
Ref.$_{{\rm 3D}}$ \\ 
& & & [dex] & & & [dex] & \\     
\hline 
 6 & C (CH)            &  &         &      &  &  $-$0.5 to $-$0.3 & a, c \\  
 7 & N (CN)            &  &         &      &  &		   $-$0.5 & see text \\  
 8 & O                 &  & $-$0.1  & i    &  &  $-$0.1	          & b \\  
11 & Na                &  & $-$0.4  & d    &  &  $-$0.1 to +0.1   & a \\  
12 & Mg                &  &   +0.1  & g    &  &  $-$0.5 to $-$0.3 & a \\  
13 & Al                &  &   +0.5  & e    &  &  $-$0.5 to $-$0.3 & a \\  
20 & Ca                &  &   +0.2  & h    &  &  $-$0.3 to $-$0.1 & a \\  
26 & Fe (\ion{Fe}{i})  &  &   +0.6  & f    &  &  $-$0.5 to $-$0.3 & a \\  
26 & Fe (\ion{Fe}{ii}) &  &   +0.2  & b    &  &  $-$0.1 to +0.1   & a \\  
38 & Sr                &  &   +0.3  & b    &  &  $-$0.5 to $-$0.3 & a \\  
56 & Ba                &  &   +0.2  & b    &  &  $-$0.5 to $-$0.1 & a \\  
63 & Eu                &  &   +0.1  & b    &  &		          &   \\  
\hline
\end{tabular}}
\smallskip
References: 
a -- Asplund~(\cite{asplund2004}),
b  --  Asplund~(\cite{asplund2005}),
c  --  Asplund \& Garc{\'i}a P{\'e}rez~(\cite{asplund2001}),
d  --  Baum{\"u}ller et~al.~(\cite{baumuller1998}),
e  --  Baum{\"u}ller \& Gehren (\cite{baumuller1997}),
f  --  Collet et~al.~(\cite{collet2005}),
g  --  Gehren et~al.~(\cite{gehren2004}),
h  --  Korn \& Mashonkina (2006, in preparation),
i  --  Nissen et~al.~(\cite{nissen2002}).
\end{table}